Math 9C Winter 2006 R. Schultz

UPDATED GENERAL INFORMATION — JANUARY 25, 2006

This is a review of information furnished in today’s lecture regarding the examination one week from today. It will cover the course material through Section 8.4, and there will be six problems. Working the homework would be good preparation for the test because 85 per cent of it will be taken directly from the homework assignments through Section 8.4 and the remaining part will involve a problem at a comparable level that is closely related to points covered in the assignments or the lectures.

If you have questions about the answers to the even numbered exercises (not in the back of the textbook), I am willing to furnish them if feasible (allow a reasonable amount of time for a response).

No books, notes or electronic devices may be used during the examination. As noted before, you should bring your student identification card in case there are identification checks.

Current plans call for the examinations to be graded and by Tuesday, February 7, and returned at the discussion sections on that date.